Author's response to reviews

Title: Validation of Administrative Health Data for the Pediatric Population: A Scoping Review

Authors:

Natalie J Shiff (natalie_shiff@hotmail.com)
Sadia Jama (sjama036@gmail.com)
Catherine Boden (catherine.boden@usask.ca)
Lisa M Lix (lisa.lix@med.umanitoba.ca)

Version: 2 Date: 16 July 2013

Author's response to reviews:

We have completed the PRISMA checklist and reformatted the submission as requested. Please note that this is a scoping review, and not a systematic review or meta-analysis, thus some of the items in the PRISMA checklist were not relevant. These items are marked NA on the checklist.